Author’s response to reviews

Title: Continuous Palliative Sedation for Patients with Advanced Cancer at a Tertiary Care Cancer Center

Authors:

Bernard Prado (pradobernard@hotmail.com)
Pedro Uson-Jr (pedroluiz_uson@hotmail.com)
Diogo Bugano (diogobugano@gmail.com)
Patricia Taranto (pate_taranto@hotmail.com)
Monique França (falecommonique@hotmail.com)
Daniel Eiger (danieleiger@gmail.com)
Rodrigo Mariano (ro_coutinho18@hotmail.com)
David Hui (dhui@mdanderson.org)
Auro Del Giglio (aurodelgiglio@gmail.com)

Version: 3 Date: 08 Dec 2017

Author’s response to reviews:

1. Please include the full name of the ethics committee (and the institute to which it belongs to) that approved the study and the committee’s reference number if appropriate, in the appropriate section of the Declarations.

Response: This was included in the Declarations section, page 12, lines 293-295.

2. Thank you for providing a statement regarding the Availability of data and materials. However, we cannot currently accept the information provided as it does not meet our editorial policies (see http://www.biomedcentral.com/getpublished/editorial-policies#DataandMaterialRelease).

Submission of a manuscript to a BioMed Central journal implies that materials described in the manuscript, including all relevant raw data, will be freely available to any scientist wishing to use them for non-commercial purposes, without breaching participant confidentiality.

Please detail where the raw data supporting your findings can be found (including information found in any supplementary files). If the raw data is publicly available or can be requested,
please state that this is the case, and where the data can be found/requested from. Alternatively, if you do not wish to/cannot share your data, please state (in the ‘Availability of data and Materials’ section) that data will not be shared, and state the reason.

Please note that participant or patient data should be de-identified.

Please see our submission guidelines for the statements we can accept in this section.

Response: A revised statement on this issue was included. Declaration section, page 12, lines 297-298.

3. Please consider the list of authors as it currently stands with reference to our guidelines regarding qualification for authorship (http://www.biomedcentral.com/submissions/editorial-policies#authorship).

Response: A revised statement of author’s contribution was included. Declarations section, page 12, line 303-305.